


o The filter change mock-up is planned for April 15 and 16. NWP has completed a critical review the filter
replacement schedule and determined that they can be task ready on 5/12 (ten days earlier). Additional filters
are expected on 4/17. (No Change) '

e The current Mod filter DPs are: 856 HEPA Bank —2.24; 857 HEPA Bank — 1.74. The alarm set point is 2.5. NWP is
evaluating making an adjustment to the Mod filter alarm point this week. Close monitoring ¢ the mod filter
dp’s is being performed while continuing to prepare for filter replacement.

e Shipments of TRU waste from LANL to WCS are continuing: 7 shipments are expected this week due to the
holiday.

NOTE CALL-IN NUMBER: 1 888 413 3490, Code 7175394
Thanks
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